Author’s response to reviews

Title: microRNA-21 promotes breast cancer proliferation and metastasis by targeting LZTFL1

Authors:

Hui Wang (027whwh@sina.com)
Zheqiong Tan (tanzheqiong@csu.edu.cn)
Hui Hu (476037893@qq.com)
Hongzhou Liu (azura@whu.edu.cn)
Tangwei Wu (wutangweiw@ sina.cn)
Chao Zhen (479332484@qq.com)
Xiuling Wang (xiuling19890430@163.com)
Zhenzhao Luo (lzzbzy@163.com)
Jing Wang (wangjing1219_2007@163.com)
Shuiyi Liu (13657299903@139.com)
Zhongxin Lu (lzx71@yahoo.com)
Jiancheng Tu (jianchengtu@whu.edu.cn)

Version: 5 Date: 17 Jul 2019

Author’s response to reviews:

Jul 17, 2019

Dear Editors,

Thank you very much for providing us an opportunity to revise the manuscript. We appreciate reviewers for their positive comments and constructive critiques. We revised the manuscript and listed our point-to-point response to reviewers’ comments in the rebuttal letter.
We would like to submit this revised manuscript to BMC Cancer, and hope it is acceptable for publication. If you have any question, please do not hesitate to contact me at the address below. Thanks again and I am looking forward to hearing from you soon.

Sincerely,

Jiancheng Tu, Prof.
Department & Program of clinical laboratory Medicine, center for gene Diagnosis, Zhongnan hospital of Wuhan University, 169 Donghu road, Wuhan 430071, People’s republic of china
Tel +86 27 6781 2989  Fax +86 27 6781 3233
Email : jianchentu@whu.edu.cn

Point-by-point Response:

Editor Comments:

1- Please provide specific Authors' contributions for all of the authors in the Authors' contributions section by using unique initials; there are two authors with ZL as initials.

Answer: Thank you for your comment. We have corrected the two authors’ unique initials as ZZL for Zhenzhao Luo and ZXL for Zhongxin Lu in Authors' contributions Section (Authors' contributions section, line 2-5, page 19).